Author's response to reviews

Title: Factors associated with pastoral community knowledge and occurrence of mycobacterial infections in Human-Animal Interface areas of Nakasongola and Mubende districts, Uganda

Authors:

Clovice Kankya (clokankya@yahoo.com)
Adrian Muwonge (adrian.muwonge@nvh.no)
Susan Olet (susan.olet@ndsu.edu)
Musso Munyeme (munyeme@yahoo.co.uk)
Demelash Biffa (Aredares@veths.no)
John Prof Opuda-Asibo (opas_in@yahoo.com)
Eystein Prof Skjerve (Eystein.Skjerve@nvh.no)
James As. Prof Oloya (joloya2001@yahoo.co.uk)

Version: 4 Date: 24 June 2010

Author's response to reviews: see over
RESPONSE TO COMMENT FROM REVIEWERS

We humbly appreciate the comments from reviewers 1 and 2, and we would like to respond to the issue from reviewers 1 as follows:

Reviewer 1:

Major Compulsory Revision:

Query: The misrepresentation on lines 315-316.

Response: Authors’ have effectively made the change as recommended by the reviewer.

Reviewer 2:

Major revision: None.

Minor revision: None

Discretionary revisions: None

Thank you